Author’s response to reviews

Title: Determinants Of Poor Glycemic Control Among Adult Patients With Type 2 Diabetes Mellitus In Jimma University Medical Center, Jimma Zone, South West Ethiopia: A Case Control Study

Authors:

Yitagesu Mamo (mamoyitagesu@gmail.com)

Fekede Bekele (fekedeb@gmail.com)

Tadesse Nigussie (tadessenigussie21@gmail.com)

Ameha Zewudie (amehazewudie@gmail.com)

Version: 1 Date: 05 Apr 2019

Author’s response to reviews:

I have tried to respond to all comments and questions raised. I wrote the response for each comment separately on the Response letter attached.